# Primary care-based screening for cardiovascular risk factors in patients with psoriasis\*

M.K. Rutter,<sup>1,2</sup> K. Kane,<sup>3</sup> M. Lunt,<sup>4,5</sup> L. Cordingley,<sup>3,6</sup> A. Littlewood,<sup>3</sup> H.S. Young,<sup>3,7</sup> C.A. Chew-Graham,<sup>8,9</sup> R. Hilton,<sup>10</sup> D.P.M. Symmons<sup>4,5</sup> and C.E.M. Griffiths<sup>3,7</sup>

<sup>1</sup>The Endocrinology and Diabetes Research Group, Institute of Human Development, Faculty of Medical and Human Sciences; <sup>2</sup>Manchester Diabetes Centre, Central Manchester University Hospitals NHS Foundation Trust; <sup>3</sup>Centre for Dermatology Research, Institute of Inflammation and Repair; <sup>4</sup>Arthritis Research U.K. Centre for Epidemiology, Centre for Musculoskeletal Research, Institute of Inflammation and Repair; <sup>6</sup>Manchester Centre for Health Psychology, School of Psychological Sciences, University of Manchester; and <sup>7</sup>Centre for Dermatology Research, Salford Royal NHS Foundation Trust; Manchester Academic Health Science Centre, Manchester, U.K.

<sup>5</sup>NIHR Manchester Musculoskeletal Biomedical Research Unit, Manchester, U.K.

<sup>8</sup>Research Institute, Primary Care and Health Sciences, Keele University, Keele, U.K.

<sup>9</sup>South Staffordshire and Shropshire Healthcare NHS Foundation Trust, Stafford, U.K.

Summary

<sup>10</sup>Bridgewater Community Healthcare NHS Trust, Wigan, U.K.

# Correspondence

Martin K. Rutter. E-mail: martin.rutter@manchester.ac.uk

### Accepted for publication

24 February 2016

### **Funding sources**

The Identification and Management of Psoriasis Associated ComorbidiTy (IMPACT) study was funded by funded by the National Institute for Health Research under its Programme Grants for Applied Research scheme (RP-PG-0608-10163).

#### **Conflicts of interest**

None declared.

\*Plain language summary available online

DOI 10.1111/bjd.14557

Background Studies assessing cardiovascular disease (CVD) risk factors in patients with psoriasis have been limited by selection bias, inappropriate controls or a reliance on data collected for clinical reasons.

Objectives To investigate whether screening for CVD risk factors in patients with psoriasis in primary care augments the known prevalence of CVD risk factors in a cross-sectional study.

Methods Patients listed as having psoriasis in primary care were recruited, screened and risk assessed by QRISK2.

Results In total, 287 patients attended (mean age 53 years, 57% women, 94% white British, 22% severe disease, 33% self-reported psoriatic arthritis). The proportion with known and screen-detected (previously unknown) risk factors was as follows: hypertension 35% known and 13% screen-detected; hypercholestero-laemia 32% and 37%; diabetes 6.6% and 3.1% and chronic kidney disease 1.1% and 4.5%. At least one screen-detected risk factor was found in 48% and two or more risk factors were found in 21% of patients. One in three patients (37%) not previously known to be at high risk were found to have a high (> 10%) 10-year CVD risk. Among the participants receiving treatment for known CVD risk factors, nearly half had suboptimal levels for blood pressure (46%) and cholesterol (46%).

Conclusions Cardiovascular risk factor screening of primary care-based adults with psoriasis identified a high proportion of patients (i) at high CVD risk, (ii) with screen-detected risk factors and (iii) with suboptimally managed known risk factors. These findings need to be considered alongside reports that detected limited responses of clinicians to identified risk factors before universal CVD screening can be recommended.

### What's already known about this topic?

- Several studies have suggested that patients with psoriasis have a greater number of risk factors for cardiovascular disease (CVD) and a higher risk for vascular events compared with the general population.
- However, the prevalence data used to support systematic CVD risk factor screening in psoriasis have been limited by selection bias, inappropriate choice of control groups or reliance on risk factors measured for other clinical reasons.

**348** British Journal of Dermatology (2016) **175**, pp348–356

© 2016 The Authors. British Journal of Dermatology

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# What does this study add?

- Cardiovascular risk factor screening of primary care-based adults with psoriasis identified a high proportion of patients (i) at high CVD risk, (ii) with screen-detected risk factors and (iii) with suboptimally managed known risk factors.
- These findings need to be considered alongside reports that detected limited responses of clinicians to identified risk factors before universal CVD screening can be recommended.

Several studies have suggested that patients with psoriasis have a greater number of risk factors for cardiovascular disease (CVD) and a higher risk for vascular events compared with the general population.<sup>1</sup> However, the prevalence data used to support systematic CVD risk factor screening in psoriasis have been limited by selection bias, inappropriate choice of control groups or reliance on risk factors measured for other clinical reasons.<sup>1–4</sup>

No previous primary care-based study has assessed how CVD risk factor screening influences the estimated prevalence of CVD risk factors in patients with psoriasis or estimated the potential benefits of CVD risk factor identification or intervention.

Our aims were (i) to investigate whether CVD risk factor screening in primary care could influence the estimated prevalence of CVD risk factors in patients with psoriasis, (ii) to assess whether the prevalence of screen-detected CVD risk factors or CVD risk varies by age, psoriasis severity and the presence of psoriatic arthritis (PsA) and (iii) to calculate the expected clinical benefits of normalizing modifiable risk factors.

# Materials and methods

### Selection of general practices

We recruited patients from 13 general practices within North West England sampled on list size and deprivation level based on practice postcode.

### Selection of participants

Participating practices were invited to identify adults with psoriasis using Read codes mapping to psoriasis and current prescribing of psoriasis medications (Appendix 1). Exclusion criteria were severe mental health problems, those without capacity to consent, recently bereaved and terminally ill individuals. Eligible patients with psoriasis were sent an invitation by their general practitioner (with a reminder when appropriate) to attend their practice for CVD risk screening.

### **Data collection**

Respondents were asked to self-complete several questionnaires before the assessment visit [Self-assessment Psoriasis Area and Severity Index (PASI),<sup>5</sup> Rose Angina Questionnaire,<sup>6</sup> Psoriasis Epidemiology Screening Tool]<sup>7</sup> and during the visit (PsA questions, medical history, psoriasis history, social and family history, therapy and comorbidity) with some additional face-to-face questions asked by the nurse (alcohol and smoking). Examination included height, weight (Seca 877; Seca, Hamburg, Germany) and blood pressure (Omron 907 BPM; Omron, Osaka, Japan). A fasting blood sample was taken for glycated haemoglobin (HbA<sub>1c</sub>), lipids, glucose, liver and renal function.

### Known cardiovascular disease risk factors

Practice staff provided information about known CVD risk factors. We used this information along with information about medication and self-reported risk factors to assess whether risk factors were newly detected by screening.

### **Comparison with Health Survey for England**

The prevalence of risk factors was compared with that in a general population sample drawn from the Health Survey for England 2011. $^{8}$ 

### Exposure and outcome definitions

'PsA' was defined as positive responses to any three out of the five Psoriasis Epidemiology Screening Tool questions<sup>7</sup> or a positive response to the question 'Have you ever been told that you have arthritis associated with psoriasis?'.

'Severe psoriasis' was defined as a self-assessment PASI score > 10 or currently receiving one or more psoriasis medications (Appendix 2).

We used the following definitions for abnormal values also used in the Health Survey for England.

'Hypercholesterolaemia' was defined as a positive response to the question 'Have you been told you have high cholesterol or high blood fat levels?', the use of lipid-lowering therapy or total cholesterol > 5 mmol L<sup>-1</sup>.

'Diabetes' was defined as a positive response to the question 'Have you ever been told by a doctor you have diabetes?' or  $HbA_{1c} \geq 48 \ mmol \ mol^{-1}$  or the use of diabetes medication.

'Hypertension' was defined as a positive response to the question 'Have you been told you have high blood pressure?', systolic blood pressure > 140 mmHg, diastolic blood pressure > 90 mmHg or use of antihypertensive medication.

'Alcohol excess' was defined as > 21 units per week in men and > 14 units per week in women.

'Suboptimal levels of risk factors on therapy' were defined as systolic blood pressure: > 140 mmHg, diastolic blood pressure > 90 mmHg, cholesterol > 5 mmol  $L^{-1}$  and HbA<sub>1c</sub>  $\geq$  53 mmol mol<sup>-1</sup>.

'Rheumatoid arthritis' was defined as a positive response to the question 'Have you ever been told by a doctor that you have rheumatoid arthritis?'.

### Sample size calculations

We aimed to recruit 320 people to yield estimates of the prevalence of CVD risk factors to within ~ 1.9% of the true value for rarer factors and to within ~ 5.3% of the true value for common factors.

### Statistical methods

Data are reported as mean (SD) or median (range) depending on data distribution. Groups were compared using  $\chi^2$ -test or Fisher's exact test for categorical data and Student's t-test or the Mann–Whitney U-test for continuous data.

Sampling weights for the Health Survey for England data were calculated by sex, ethnicity and 5-year age band, weights being equal to the number of subjects in the stratum in the psoriasis sample divided by the number of subjects in that stratum in the Health Survey for England data. These weights were used as sampling weights to standardize Health Survey for England estimates to the age, ethnicity and sex distribution of the psoriasis sample.

QRISK2 data were calculated by using STATA plugins using the QRISK-2013 source code (ClinRisk Ltd., Nottingham, U.K.). Using the QRISK2 calculator, the optimized CVD risk was calculated using an ideal set of modifiable risk factors for each individual (no smoking, cholesterol/high-density lipoprotein ratio = 5, systolic blood pressure = 140 mmHg and body mass index = 25 kg m<sup>-2</sup>). The absolute change in predicted risk through risk factor optimization was calculated as the optimized risk minus the predicted risk before risk factor optimization.

P < 0.05 was considered statistically significant. All analyses were performed using STATA 13 (StataCorp, College Station, TX, U.S.A.).

### **Research ethics**

The study was approved by the local research ethics committee [North West Research Committee, Greater Manchester East (REC ref: 11/NW/0654)]. Written informed consent was obtained from all participants.

### Results

Invitations to attend were sent to 1446 patients, 287 of whom (20%) enrolled. Overall, 22% of participants had severe psoriasis defined by a high self-administered PASI score (7.4% had an index of > 10) or use of disease-modifying therapies (16% of participants) and one-third self-reported PsA (Table 1).

More than one-third (35%) of patients were clinically obese, 52% had a very high waist circumference (> 102 cm for men; > 88 cm for women) and one in six (18%) participants were current smokers (Table 1). Diabetes was self-reported in 6-6% and two-thirds of these patients were receiving diabetes medication. Approximately one-third selfreported high cholesterol levels (29%) and high blood pressure (33%) and over two-thirds of these were receiving medication for these conditions [21% of the total cohort were taking lipid-lowering medication (96% were taking a statin); 25% were taking antihypertensive medication].

A self-reported history of coronary or cerebrovascular disease or atrial fibrillation was reported by less than 10% of participants (Table 1). In a separate assessment, the Rose Angina Questionnaire elicited a history of myocardial infarction in 7.7% of participants, angina in 7.7% and intermittent claudication in 1.4%.

# Prevalence of known and screen-detected (not previously known) cardiovascular disease risk factors

The proportion of participants with known and screendetected (not known) CVD risk factors was as follows: hypertension, 35% known and 13% screen-detected; hypercholesterolaemia, 32% and 37%; diabetes, 6.6% and 3.1%; and chronic kidney disease, 1.1% and 4.5% (Fig. 1). At least one screen-detected risk factor was found in 48% of patients and at least two risk factors were found in 21% of patients. The prevalence of screen-detected risk factors did not vary significantly by age, psoriasis severity or by the presence of selfreported PsA (Table S1; see Supporting Information).

# Comparison of prevalent cardiovascular disease risk factors with Health Survey for England data

In an age-, sex- and ethnicity-weighted analysis, the prevalence of hypertension was 12% higher (P < 0.001) in study participants and the prevalence of hypercholesterolaemia was 7% lower (P = 0.03; Fig. 2) than in the general population.

As ciclosporin can cause hypertension we excluded patients treated with this drug but the results were essentially unchanged [n = 269, proportion with hypertension = 47.6%, prevalence difference compared with Health Survey for England data = 11.2%; 95% confidence interval (4.6-17.8)].

Some,<sup>9–11</sup> but not all,<sup>12</sup> previous studies have suggested that methotrexate therapy is associated with a lower serum cholesterol. In our study cohort, prevalent

Table 1	Clinical ch	aracteristics	of Identi	ification	and	Mana	agement	of
Psoriasis	Associated	Comorbidi	Гу study	particip	ants,	n =	287	

Variable	Data
Age, years	53 (15)
Female sex	57
Index of multiple deprivation	19 (10-49)
for practice [median (range)]	
Ethnicity	
White British	94
Other white	3
Asian/Asian British	3
Weight, kg	81 (17)
BMI, kg m <sup>-2</sup>	28.8 (5.9)
Obesity (BMI $\ge$ 30 kg m <sup>-2</sup> )	35
Waist circumference, cm	96 (15)
Smoking	
Current	18
Previous	39
Family history of premature	19
cardiovascular disease	
Psoriasis severity and psoriatic arthriti	is
Severe psoriasis <sup>a</sup>	22
Self-administered PASI, median	3.4 (0-26.9); 4.2 (3.8
(range); mean (SD)	
Self-reported psoriatic arthritis	33
Self-reported cardiovascular disease ri	sk factors
Hypercholesterolaemia	29
Hypertension	33
Diabetes	6.6
Cardiovascular disease <sup>b</sup>	8.4
Chronic kidney disease	1.1
Rheumatoid arthritis	7.7
Atrial fibrillation	5.6
Screening test results (all patients)	
Total cholesterol, mmol L <sup>-1</sup>	5.1 (1.2)
LDL cholesterol, mmol $L^{-1}$	3.1 (1.0)
HDL cholesterol, mmol $L^{-1}$	1.5 (0.4)
Non-HDL cholesterol, mmol $L^{-1}$	3.2 (1.2)
Triglycerides, mmol $L^{-1}$ ,	1.3 (0.4-4.9)
median (range)	
Blood pressure, mmHg	130 (18)/78 (11)
Fasting plasma glucose, mmol L <sup>-1</sup>	5.6 (5.3)
$HbA_{1c} \text{ mmol mol}^{-1}$	38.3 (7.1)
eGFR. mL min $^{-1}$	89 (20)

BMI, body mass index; PASI, Psoriasis Area and Severity Index; LDL, low-density lipoprotein; HDL, high-density lipoprotein; eGFR, estimated glomerular filtration rate; HbA<sub>1c</sub>, glycated haemoglobin. Data are provided as percentages or mean (SD) values, unless otherwise stated. <sup>a</sup>Defined as Self-administered PASI > 10 or use of disease-modifying therapy (psoralen–ultraviolet A, methotrexate, ciclosporin, acitretin, fumaric acid esters, etanercept, adalimumab, infliximab or ustekinumab). <sup>b</sup>Self-reported myocardial infarction, stroke, transient ischaemic attack, coronary artery bypass surgery, coronary angioplasty or carotid endarterectomy.

hypercholesterolaemia was lower in subjects taking methotrexate compared with those not taking methotrexate (59% vs. 70%, P-value not significant). When we excluded patients taking methotrexate from the psoriasis cohort, the prevalence of hypercholesterolaemia was not statistically different from the prevalence in the Health Survey for England (P = 0.08).

# Level of control of risk factors in participants with known risk factors

Among the study participants receiving treatment for CVD risk factors, nearly half had suboptimal levels for blood pressure (46%) and/or total cholesterol (46%) and one-quarter of participants had suboptimal levels for glucose assessed by HbA<sub>1c</sub> (26%) (Table S2; see Supporting Information).

# Predicted 10-year risk and the anticipated response to risk factor optimization

Using current National Institute for Health and Care Excellence (NICE) guidelines, most patients with established CVD, type 1 diabetes, familial hypercholesterolaemia or renal impairment are considered to be at high risk of CVD and unsuitable for CVD risk calculation.<sup>13</sup> In our study, participants not considered 'high risk' by NICE guidelines had an average QRISK2-predicted 10-year CVD risk of 10%, meaning that over 10 years 10% would be expected to have a CVD event (Table 2). Owing to the skewed risk distribution, just over one-third of these patients (37%) were assessed to be at high CVD risk using the 10-year CVD risk threshold of > 10%.<sup>13</sup> The predicted average absolute reduction in 10-year risk achieved through optimizing modifiable risk factors was 1.6%, giving a number needed to treat (NNT) of 60 patients to prevent one vascular event over 10 years.

# Predicted lifetime risk and the anticipated response to risk factor optimization

The mean predicted lifetime CVD risk estimated by QRISK2 was 35%. Overall, 13% of participants had a high lifetime CVD risk of > 50%. Optimization of modifiable risk factors would reduce the lifetime risk by 2.8%, giving an NNT of 36 patients to prevent one vascular event over a lifetime.

# Predicted risk and response to risk factor optimization by psoriasis severity and psoriatic arthritis

Compared with patients without self-reported PsA, those with self-reported PsA had a 77% higher predicted 10-year CVD risk assessed by QRISK2 (Table 2; 14.0% vs. 7.9%, P = 0.001). This higher risk largely disappeared after adjusting for mean age, which was higher in those with self-reported PsA (58 years vs. 51 years; analysis not shown).

The absolute reduction in 10-year and lifetime CVD risk, predicted through risk factor optimization, was higher for participants with severe psoriasis and self-reported PsA compared with those without these conditions (Table 2).

In participants < 40 years of age, the average CVD risks were much lower [10-year CVD risk = 0.7%, lifetime CVD

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists.



Fig 1. Proportion of Identification and Management of Psoriasis Associated ComorbidiTy (IMPACT) study participants with known and screendetected cardiovascular disease risk factors. Data are proportions of IMPACT study participants. 'Known' is the sum of (i) self-report, (ii) medical or nursing staff knowing about this risk factor and (iii) medication for this risk factor. High blood pressure was defined as systolic blood pressure  $\geq 140$  mmHg or diastolic blood pressure  $\geq 90$  mmHg. High cholesterol was defined as total cholesterol  $\geq 5$  mmol L<sup>-1</sup>. Diabetes was defined by high glycated haemoglobin  $\geq 48$  mmol mol<sup>-1</sup>. Chronic kidney disease was defined by low estimated glomerular filtration rate  $\leq 60$  mL min<sup>-1</sup>.



Fig 2. Prevalence of combined known and screen-detected cardiovascular risk factors in Identification and Management of Psoriasis Associated ComorbidiTy (IMPACT) study and Health Survey for England participants. Data are proportions (95% confidence interval) of IMPACT study and Health Survey for England participants with suboptimal risk factor status. Statistical significance for between-cohort comparisons is denoted by \*P < 0.05; \*\*P < 0.001. Prevalence of high cholesterol was not significant (P = 0.08) after excluding IMPACT study patients with psoriasis taking methotrexate. High cholesterol was defined as self-reported hypercholesterolaemia or lipid-lowering therapy or total cholesterol > 5 mmol L<sup>-1</sup>. High blood pressure was defined as self-reported hypertension or use of antihypertensive therapy or systolic blood pressure  $\geq$  140 mmHg or diastolic blood pressure  $\geq$  90 mmHg. High glycated haemoglobin (HbA<sub>1c</sub>) was defined as self-reported diabetes or diabetes therapy or HbA<sub>1c</sub>  $\geq$  48 mmol mol<sup>-1</sup>. Obesity was defined as body mass index  $\geq$  30 kg m<sup>-2</sup>. Smoking was defined as current use. Alcohol excess was defined as > 21 units per week for men and > 14 units per week for women.

risk = 31%, proportion at high risk over 10 years = 0% and proportion at high risk over a lifetime = 4% (Table S3; see Supporting Information)] and the NNTs were considerably higher.

## Discussion

### Main findings

Screening patients with psoriasis recruited from primary care showed that 37% of those not defined by NICE to be at 'high

risk' (such as those with prior myocardial infarction) had a high 10-year CVD risk and that those with self-reported PsA were at highest risk, mainly because they were older. At least one screen-detected (previously unknown) CVD risk factor was found in 48% of patients. The age- and sex-adjusted prevalence of hypertension was higher than in the general population and a high proportion of patients with known CVD risk factors had suboptimal control of those risk factors. Finally, we showed that optimization of modifiable CVD risk factors would be predicted to yield clinically relevant reductions in CVD event rates particularly in those with self-reported PsA. **Table 2** QRISK2-predicted 10-year and lifetime cardiovascular disease (CVD) risks by the presence of severe psoriasis and psoriatic arthropathy before and after theoretical CVD risk factor optimization in participants not considered to be high risk by National Institute for Health and Care Excellence criteria<sup>a</sup>

		Severe psoriasis			Psoriatic arthropathy		
	All patients n = 221	Yes n = 47	No n = 174	P-value	Yes n = 76	No n = 145	P-value
10-year risk							
Proportion with high 10-year CVD risk (> 10%)	37	28	39	0.15	51	29	0.001
Proportion with high 10-year CVD risk (> 20%)	17	17	15	0.7	26	12	0.006
Mean (SD) risk before risk factor optimization	10.0 (11.3)	8.9 (11.5)	10.3 (11.3)	0.5	14.0 (13.8)	7.9 (9.1)	0.001
Mean (SD) risk after risk factor optimization	8.4 (10.0)	7.2 (10.1)	8.7 (10.0)	0.4	11.8 (12.2)	6.6 (8.1)	0.001
Absolute change in predicted risk through	-1.6	-1.8	-1.6		-2.2	-1.4	
risk factor optimization, absolute percentage							
Estimated number needed to treat to prevent one CVD event over 10 years	61	57	61		45	73	
Lifetime risk							
Proportion with lifetime risk $> 50\%$	13	15	12	0.6	17	10	0.15
Mean (SD) risk before risk factor optimization	35.2 (13.3)	38.7 (16.1)	34.2 (12.2)	0.1	36.6 (14.9)	34.4 (12.3)	0.4
Mean (SD) risk after risk factor optimization	32.3 (11.7)	35.0 (14.0)	31.6 (10.9)	0.13	33.4 (13.4)	31.8 (10.7)	0.6
Absolute change in predicted risk through risk factor optimization, absolute percentage	-2.8	-3.7	-2.6		-3.2	-2.6	
Estimated number needed to treat to prevent one CVD event over a lifetime	36	26	38		31	38	

Data are the proportion (%) of people predicted to develop myocardial infarction, angina, stroke or transient ischaemic attack over 10 years (upper section of the table) over a lifetime (lower section of the table) unless stated. Lifetime risk is projected to age 95 years. Optimization of risk factors: absence of smoking, cholesterol/high-density lipoprotein ratio = 5, systolic blood pressure = 140 mmHg and body mass index = 25. Improvement in CVD risk is determined using the QRISK2 website calculators http://qrisk.org and http://www.qrisk.org/life-time/. Absolute change in predicted risk through risk factor optimization has been calculated as predicted risk after risk factor optimization minus predicted risk before risk factor optimization. Number needed to treat has been calculated as 100 divided by the absolute percentage risk reduction. Severe psoriasis was defined as patients with Psoriasis Area and Severity Index score > 10, and those receiving psoralen–ultraviolet A, methotrexate, ciclosporin, acitretin, fumaric acid esters, etanercept, adalimumab, infliximab or ustekinumab. Psoriatic arthropathy was defined as positive responses to any three of the first five Psoriasis Epidemiology Screening Tool questions<sup>7</sup> or a positive response to the question 'Have you ever been told that you have arthritis associated with psoriasis?'. "Patients with self-reported myocardial infarction, stroke, transient ischaemic attack, coronary artery bypass surgery, coronary angioplasty, carotid endarterectomy, estimated glomerular filtration rate < 60 mL min<sup>-1</sup>, familial hypercholesterolaemia or type 1 diabetes mellitus were excluded from the analysis.

#### Prior population-based and primary care-based studies

This is the first primary care-based study reporting that CVD risk factor screening augments the estimated prevalence of CVD risk factors in psoriasis. It is the first to assess lifetime CVD risk and the potential benefits of optimizing modifiable CVD risk factors.

The only previous European population-based CVD risk factor prevalence study in psoriasis, was part of the larger Rotterdam study. It involved 262 participants with psoriasis aged over 55 years (mean age 64 years) recruited from a suburb of Rotterdam.<sup>14</sup> Those with psoriasis had higher smoking prevalence (32% vs. 22%), higher body mass index (27·1 kg m<sup>-2</sup> vs. 26·5 kg m<sup>-2</sup>), higher diastolic blood pressure and greater use of lipid-lowering medication compared with study participants without psoriasis.

Three population-based studies from the U.S. National Health and Nutrition Examination Survey (NHANES) compared the prevalence of selected CVD risk factors in people with and without psoriasis.<sup>15–17</sup> One of these studies showed that psoriasis was linked to a higher likelihood of prevalent metabolic syndrome (driven by obesity and hypertension) but with no higher risk for dyslipidaemia or diabetes, which was largely confirmed in subsequent studies focusing on individual risk factors.<sup>16,17</sup>

### Prior studies in highly selected cohorts

A small number of studies have reported the prevalence of screen-detected CVD risk factors in patients with psoriasis in highly selected patient groups.<sup>18–21</sup> These studies included trial participants,<sup>20</sup> disease registry participants<sup>18</sup> and hospitalized patients,<sup>19,21</sup> who are unlikely to be representative of patients recruited from primary care.

Several large studies, some of which had been populationbased studies,<sup>22–25</sup> have reported the prevalence of known CVD risk factors.<sup>22,24–30</sup> When compared with control populations (see below), patients with psoriasis had a higher likelihood of hypertension,<sup>22,25–28,31,32</sup> dyslipidaemia,<sup>22,25,28,29,31,32</sup> diabetes,<sup>22,23,25,26,28,31,32</sup> obesity<sup>22,26,31</sup> smoking,<sup>22,26,32</sup> renal disease<sup>23,25,28</sup> and metabolic syndrome.<sup>24,28,30</sup> However, control groups in these studies included people with 'forms of dermatitis',<sup>26</sup> other hospitalized patients<sup>21,28</sup> and people taking part in other research projects,<sup>21</sup> which makes interpretation, comparison and clinical implications uncertain.

Several other large studies have simply reported the prevalence of known CVD risk factors in patients with psoriasis without a control group.<sup>33,34</sup> While these studies have provided valuable data, they do not inform about the value of primary care screening. Our study demonstrated that using a standard approach to CVD screening of patients with psoriasis who attend primary care is an effective means of identifying individuals with CVD risk factors who would have been otherwise missed.

#### Age, psoriasis severity and psoriatic arthritis

We showed that the proportion of patients with screendetected CVD risk factors was unrelated to age, psoriasis severity or the presence of self-reported PsA, suggesting that these factors are unlikely to be clinically useful in increasing the yield of abnormal risk factors from screening.

However, compared with patients without self-reported PsA, those reporting PsA had a higher 10-year CVD risk largely because they were older. For this reason, older patients, including those with self-reported PsA, may particularly benefit from CVD screening and risk factor intervention.

#### Comparison with Health Survey for England data

When known and screen-detected risk factors were considered, hypertension was more prevalent in patients with psoriasis, which is in keeping with data from the Rotterdam study and NHANES.<sup>14,17</sup>

We showed a lower prevalence of hypercholesterolaemia in patients with psoriasis than in the general population that could be explained by use of methotrexate. Even if the prevalence of hypercholesterolaemia is lower in patients with psoriasis than in the general population, this might not diminish the importance of the condition because more than one-third of these patients had screen-detected hypercholesterolaemia.

Although we had an adequate sample size, our results did not show statistically significant differences in the prevalence of obesity, smoking or diabetes compared with the general population.

#### Suboptimal management of known risk factors

We observed many patients with suboptimal management of known risk factors. Previous studies have shown undertreatment of CVD risk factors in patients with psoriasis<sup>20,35</sup> and one study suggested that hypertension is more difficult to manage in these patients,<sup>29</sup> perhaps because psoriasis or its treatment somehow increases blood pressure or reduces compliance with therapy.

# Estimated lifetime cardiovascular disease risk and the benefits of intervention

We considered lifetime CVD risk because younger high-risk patients could be missed out using 10-year risk estimates, as age dominates risk calculation. However, after excluding those automatically designated as high risk, we found that a high lifetime risk was present in only one in eight (13%) participants overall [and only one in 25 (4%) of those aged < 40 years]. However, more than one in three participants (37%) were at high 10-year CVD risk. The majority (20 of 28, 71%) of those at high lifetime risk would be detected by having a high (> 10%) 10-year CVD risk. Therefore, the estimation of lifetime risk does not appear to be particularly beneficial in patients with psoriasis.

### **Risk identification and risk reduction**

'Health checks' have recently come under criticism for failing to improve long-term health outcomes,<sup>36,37</sup> perhaps because of overmedication or social class biases in those attending.<sup>38</sup> An alternative explanation has been made by our research team in a parallel study that recorded and analysed CVD consultations. They found that risk information was seldom discussed with patients and instead practitioners prioritized information collection.<sup>39</sup> Thus, while the current study has shown that screening identifies modifiable CVD risk factors, caution is required before recommending this as a strategy.

### Strengths and limitations

Strengths of this study include the following: (i) we reported how CVD risk factor screening influenced the detected prevalence of CVD risk factors; (ii) our data can inform policy on primary care-based CVD screening; (iii) psoriasis severity was assessed; (iv) the prevalence of important CVD risk factors was established with clinically meaningful precision in important subgroups; (v) we compared prevalence data with population-based controls matched for age, sex and ethnicity; and (vi) we assessed the benefits of optimizing risk factors within 10-year and lifetime time frames.

The limitations of this study include the following: (i) risk factors were measured on one occasion, which may have led to some misclassification, (ii) our cohort was largely white limiting the generalizability of findings, (iii) the participation rate was 20%; we suspect that patients who did not participate would be more likely to have an adverse risk factor profile than those who enrolled, and (iv) we did not validate diagnoses of psoriasis or PsA.

#### **Clinical implications**

Cardiovascular risk factor screening is a means of identifying a high proportion of patients at high CVD risk, with screendetected risk factors and with suboptimally managed known risk factors. However, these findings need to be considered alongside those showing the limited response of clinicians to identified risk before universal CVD screening for people with psoriasis can be recommended.<sup>39</sup>

### Conclusion

Screening for CVD risk factors in patients with psoriasis identified a high proportion of high-risk individuals with potentially modifiable CVD risk factors in addition to a high proportion with suboptimally managed known risk factors. These data provide valuable augmenting information about the prevalence of CVD risk factors and the potential value of primary carebased screening of patients with psoriasis.

### Acknowledgments

This article presents independent research funded by National Institute of Health Research (NIHR) under its Programme Grants for Applied Research scheme (RP-PG-0608-10163). The views expressed are those of the authors and not necessarily those of the National Health Service, the NIHR or the Department of Health. The authors would like to thank the patients and practitioners who took part in this study and to acknowledge the support of the NIHR through the Comprehensive Clinical Network for recruitment. The study was facilitated by the Manchester Centre for Dermatology Research, Arthritis Research U.K. Centre for Epidemiology, Manchester Biomedical Research Centre and Manchester Academic Health Science Centre. M.K.R. and H.S.Y. were supported by the Higher Education Funding Council for England (Clinical Senior Lecturer Awards). D.P.M.S. and C.E.M.G. are NIHR senior investigators. C.C.-G. is funded in part by the NIHR Collaborations for Leadership in Applied Health Research and Care West Midlands.

### References

- 1 Miller IM, Ellervik C, Yazdanyar S, Jemec GB. Meta-analysis of psoriasis, cardiovascular disease, and associated risk factors. J Am Acad Dermatol 2013; 69:1014–24.
- 2 Armstrong AW, Harskamp CT, Armstrong EJ. The association between psoriasis and obesity: a systematic review and meta-analysis of observational studies. Nutr Diabetes 2012; **2**:e54.
- 3 Patel RV, Shelling ML, Prodanovich S et al. Psoriasis and vascular disease-risk factors and outcomes: a systematic review of the literature. J Gen Intern Med 2011; 26:1036–49.
- 4 Prey S, Paul C, Bronsard V et al. Cardiovascular risk factors in patients with plaque psoriasis: a systematic review of epidemiological studies. J Eur Acad Dermatol Venereol 2010; **24**(Suppl. 2):23–30.
- 5 Feldman SR, Fleischer AB Jr, Reboussin DM et al. The self-administered psoriasis area and severity index is valid and reliable. J Invest Dermatol 1996; 106:183–6.

- 6 Rose GA. The diagnosis of ischaemic heart pain and intermittent claudication in field surveys. Bull World Health Organ 1962; 27:645–58.
- 7 Ibrahim GH, Buch MH, Lawson C et al. Evaluation of an existing screening tool for psoriatic arthritis in people with psoriasis and the development of a new instrument: the Psoriasis Epidemiology Screening Tool (PEST) questionnaire. Clin Exp Rheumatol 2009; 27:469–74.
- 8 Health Survey for England. Health Survey for England: Health, Social Care and Lifestyles. Available at http://www.hscic.gov.uk/ catalogue/PUB09300/HSE2011-Sum-bklet.pdf (last accessed 30 June 2016).
- 9 Kilic S, Emre S, Metin A et al. Effect of the systemic use of methotrexate on the oxidative stress and paraoxonase enzyme in psoriasis patients. Arch Dermatol Res 2013; 305:495–500.
- 10 Georgiadis AN, Voulgari PV, Argyropoulou MI et al. Early treatment reduces the cardiovascular risk factors in newly diagnosed rheumatoid arthritis patients. Semin Arthritis Rheum 2008; 38:13– 19.
- 11 Gisondi P, Cazzaniga S, Chimenti S et al. Metabolic abnormalities associated with initiation of systemic treatment for psoriasis: evidence from the Italian Psocare Registry. J Eur Acad Dermatol Venereol 2013; 27:e30–41.
- 12 Navarro-Millán I, Charles-Schoeman C, Yang S et al. Changes in lipoproteins associated with methotrexate or combination therapy in early rheumatoid arthritis: results from the treatment of early rheumatoid arthritis trial. Arthritis Rheum 2013; **65**:1430–8.
- 13 Rabar S, Harker M, O'Flynn N, Wierzbicki AS. Lipid modification and cardiovascular risk assessment for the primary and secondary prevention of cardiovascular disease: summary of updated NICE guidance. BMJ 2014; 349:g4356.
- 14 Dowlatshahi EA, Kavousi M, Nijsten T et al. Psoriasis is not associated with atherosclerosis and incident cardiovascular events: the Rotterdam Study. J Invest Dermatol 2013; 133:2347–54.
- 15 Love TJ, Qureshi AA, Karlson EW et al. Prevalence of the metabolic syndrome in psoriasis: results from the National Health and Nutrition Examination Survey, 2003–2006. Arch Dermatol 2011; 147:419–24.
- 16 Casagrande SS, Menke A, Cowie CC. No association between psoriasis and diabetes in the U.S. population. Diabetes Res Clin Pract 2014; 104:e58–60.
- 17 Ma C, Schupp CW, Armstrong EJ, Armstrong AW. Psoriasis and dyslipidemia: a population-based study analyzing the National Health and Nutrition Examination Survey (NHANES). J Eur Acad Dermatol Venereol 2014; 28:1109–12.
- 18 Kimball AB, Leonardi C, Stahle M et al. Demography, baseline disease characteristics, and treatment history of patients with psoriasis enrolled in a multicenter, prospective, disease-based registry (PSOLAR). Br J Dermatol 2014; 171:137–47.
- 19 Ma L, Li M, Wang H et al. High prevalence of cardiovascular risk factors in patients with moderate or severe psoriasis in northern China. Arch Dermatol Res 2014; 306:247–51.
- 20 Kimball AB, Szapary P, Mrowietz U et al. Underdiagnosis and undertreatment of cardiovascular risk factors in patients with moderate to severe psoriasis. J Am Acad Dermatol 2012; 67:76–85.
- 21 Warnecke C, Manousaridis I, Herr R et al. Cardiovascular and metabolic risk profile in German patients with moderate and severe psoriasis: a case–control study. Eur J Dermatol 2011; 21:761–70.
- 22 Neimann AL, Shin DB, Wang X et al. Prevalence of cardiovascular risk factors in patients with psoriasis. J Am Acad Dermatol 2006; 55:829–35.
- 23 Yeung H, Takeshita J, Mehta NN et al. Psoriasis severity and the prevalence of major medical comorbidity: a population-based study. JAMA Dermatol 2013; **149**:1173–9.

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists.

- 24 Langan SM, Seminara NM, Shin DB et al. Prevalence of metabolic syndrome in patients with psoriasis: a population-based study in the United Kingdom. J Invest Dermatol 2012; **132**:556–62.
- 25 Yang YW, Keller JJ, Lin HC. Medical comorbidity associated with psoriasis in adults: a population-based study. Br J Dermatol 2011; 165:1037–43.
- 26 Shapiro J, Cohen AD, Weitzman D et al. Psoriasis and cardiovascular risk factors: a case–control study on inpatients comparing psoriasis to dermatitis. J Am Acad Dermatol 2012; 66:252–8.
- 27 Shapiro J, Cohen AD, David M et al. The association between psoriasis, diabetes mellitus, and atherosclerosis in Israel: a case-control study. J Am Acad Dermatol 2007; 56:629–34.
- 28 Sommer DM, Jenisch S, Suchan M et al. Increased prevalence of the metabolic syndrome in patients with moderate to severe psoriasis. *Arch Dermatol Res* 2006; 298:321–8.
- 29 Armstrong AW, Lin SW, Chambers CJ et al. Psoriasis and hypertension severity: results from a case–control study. PLoS ONE 2011; 6: e18227.
- 30 Parodi A, Aste N, Calvieri C et al. Metabolic syndrome prevalence in psoriasis: a cross-sectional study in the Italian population. Am J Clin Dermatol 2014; 15:371–7.
- 31 Schmitt J, Ford DE. Psoriasis is independently associated with psychiatric morbidity and adverse cardiovascular risk factors, but not with cardiovascular events in a population-based sample. J Eur Acad Dermatol Venereol 2010; 24:885–92.
- 32 Prodanovich S, Kirsner RS, Kravetz JD et al. Association of psoriasis with coronary artery, cerebrovascular, and peripheral vascular diseases and mortality. Arch Dermatol 2009; 145:700–3.
- 33 Armstrong AW, Schupp C, Bebo B. Psoriasis comorbidities: results from the National Psoriasis Foundation surveys 2003 to 2011. Dermatology 2012; 225:121–6.
- 34 Mazlin MB, Chang CC, Baba R. Comorbidities associated with psoriasis – data from the Malaysian psoriasis registry. Med J Malaysia 2012; 67:518–21.
- 35 Ahlehoff O, Skov L, Gislason G et al. Pharmacological undertreatment of coronary risk factors in patients with psoriasis: observational study of the Danish nationwide registries. PLoS ONE 2012; 7: e36342.
- 36 Krogsbøll LT, Jørgensen KJ, Gøtzsche PC. Universal health checks should be abandoned. BMJ 2013; **347**:f5227.
- 37 Krogsbøll LT, Jørgensen KJ, Gronhoj Larsen C, Gøtzsche PC. General health checks in adults for reducing morbidity and mortality from disease: Cochrane systematic review and meta-analysis. BMJ 2012; 345:e7191.
- 38 Gøtzsche PC, Jørgensen KJ, Krogsbøll LT. General health checks don't work. BMJ 2014; 348:g3680.
- 39 Nelson PA, Kane K, Chisholm A et al. 'I should have taken that further' – missed opportunities during cardiovascular risk assessment in patients with psoriasis in UK primary care settings: a mixedmethods study. Health Expect 2016; DOI: 10.1111/hex.12404

### Appendix 1:

### Read codes and Read terms used to identify study participants

M161.00	Other psoriasis
M161000	Psoriasis unspecified
M161600	Guttate psoriasis
M161A00	Psoriasis palmaris
M161B00	Psoriasis plantaris
M161D00	Pustular psoriasis
M161E00	Psoriasis universalis
M161F00	Psoriasis vulgaris
M161F11	Chronic large plaque psoriasis
M161H00	Erythrodermic psoriasis
M161z00	Psoriasis NOS
M16y.00	Other psoriasis and similar disorders
M16y000	Scalp psoriasis
M16z.00	Psoriasis and similar disorders NOS
Myu3000	[X]Other psoriasis

### Appendix 2:

# Medication used to identify patients with severe psoriasis

Psoralen and ultraviolet A therapy, methotrexate, ciclosporin, acitretin, fumaric acid esters, etanercept, adalimumab, infliximab or ustekinumab.

### **Supporting Information**

Additional Supporting Information may be found in the online version of this article at the publisher's website:

**Table S1.** Proportion (%) of patients with screen-detected abnormal risk factors by age, psoriasis severity and the presence of psoriatic arthritis.

**Table S2.** Proportion of Identification and Management of Psoriasis Associated ComorbidiTy study participants with known abnormal risk factors (on treatment) that fail to achieve target levels for risk factors.

**Table S3.** Characteristics of included randomized controlled trials and summary of results.

Video S1. Author Video.